

**Title: A COVID-19 patient with an incubation period of 27 days**

**NO:** 60361

Dear editors,

We thank you for your careful consideration of our manuscript. We appreciate your response and overall positive initial feedback and made modifications to improve the manuscript. After carefully reviewing the comments made by the Reviewers, we have modified the manuscript to improve it.

We hope that you will find the revised paper suitable for publication, and we look forward to contributing to your journal. Please do not hesitate to contact us with other questions or concerns regarding the manuscript.

Best regards

Yours sincerely,

Ming-Yan Zhao,

Department of Critical Care Medicine, The First Affiliated Hospital of Harbin Medical University, No. 23 Youzheng Street, Harbin 150001, Heilongjiang Province, China.

E-mail: mingyan0927@yeah.net

Telephone: +86-13796118989

**Peer-review report(s):**

Conclusion: Major revision

Scientific Quality: Grade B (Very good)

Language Quality: Grade A (Priority publishing)

Specific Comments to Authors:General Comment: Interesting case report regarding Cover-19 infection considering the atypical clinical presentation of the infection. The case is well described, minor changes are needed. Specific Comments: - Introduction: the third sentence regarding vaccines should be modified since currently vaccines are available. - Discussion: in regards of atypical Coved-19 cases, additionally to CT scan, also a reference about PET/CT imaging should be inserted in the manuscript since the Coved-19 diagnosis may be also incidental and unknown in patients undergone PET/TC imaging; the following reference could be appropriate: FDG-PET/CT imaging during the Covid-19 emergency: a southern Italian perspective. Eur J Nucl Med Mol Imaging. 2020 Oct;47(11):2691-2697. doi: 10.1007/s00259-020-04931-7. Epub 2020 Jun 23. PMID: 32572561; PMCID: PMC7308109.

**Response:** *Thanks for your comment and your positive feed back for our manuscript. We have made some modification in the conclusion and checked the manuscript again for to correct any typos we have found. We've cut the vaccine treatment in the introduction and added the diagnostic value of PET/CT in the discussion. We have highlighted them in red.*